Author’s response to reviews

Title: Preventive Evidence into Practice: What factors matter in a facilitation intervention to prevent vascular disease in family practice?

Authors:
Grant Russell (grant.russell@monash.edu)
Riki Lane (riki.lane@monash.edu)
Sharon Parker (sharon.parker@unsw.edu.au)
John Litt (jlitt@flinders.edu.au)
Danielle Mazza (danielle.mazza@monash.edu)
Jane Lloyd (j.lloyd@unsw.edu.au)
Nicholas Zwar (nzwar@uow.edu.au)
Mieke van Driel (m.vandriel@uq.edu.au)
Chris Del Mar (cdelmar@bond.edu.au)
Jane Smith (jsmith@bond.edu.au)
Mark Harris (m.f.harris@unsw.edu.au)

Version: 2  Date: 01 Jul 2019

Author’s response to reviews:

This amendment addresses the journals request as of email sent under the header FAMP-D-18-00265R2 - Manuscript Sent Back. of the 25th June 2019

Thank you for the recent correspondence regarding this manuscript. An amended version has been uploaded together with a supplementary file containing two appendices.

IN response to the specific requests:

1) we have uploaded clean versions without line numbers or highlighted text
2). Title page
Funding information had already been removed from the title page

3. Collaborators
Our collaborators have been merged into the acknowledgements subsection.

4. Additional files
Other files have been deleted.

NOW IN ADDITIONAL INFORMATION REQUESTED IN AN EMAIL OF 1 JULY 2019
FORMATTING CHANGES:

THANK YOU FOR YOUR REQUEST FOR FURTHER CHANGES IN THE MANUSCRIPT - SINCE THEY DO NOT APPEAR IN THE SYSTEM I HAVE COPIED BELOW

YOUR REQUEST:
1. Please include Consent for Publication.
OUR RESPONSE:
We have written the words "not applicable" under the section consent for publication on page 14

YOUR REQUEST:
2. Please Re-arrange the Declarations Sub-sections.
OUR RESPONSE:
the declarations sub sections have been rearranged to fit the format of the journal. These can be found on pages 14-16 of the already uploaded submission.

DECLARATIONS SECTION:

YOUR REQUEST:
2. Please Re-arrange the Declarations Sub-sections.
OUR RESPONSE
These have been rearranged

YOUR REQUEST:
2.1) For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found.
OUR RESPONSE
We have referred the editorial advice. Our original offering of available data was incorrect. We have reprinted the description used in the companion quantitative study - also published in the BMC stable (Harris et al. BMC Health Services Research (2017) 17:637 DOI 10.1186/s12913-017-2586-4): It also pertains to this study: The text reads:
"The datasets generated and/or analysed during the current study are not publicly available as they contain identifiable patient information. Please contact the first author."

This included text is on p 14